

Integrating psychosocial variables and societal diversity in epidemic models for predicting COVID-19 transmission dynamics

Viktor Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Jan Fousek, Cornelia Betsch, Lisa Felgendreff, Robert Bohm, Lau Lilleholt, Ingo

Zettler, et al.

► To cite this version:

Viktor Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Jan Fousek, et al.. Integrating psychosocial variables and societal diversity in epidemic models for predicting COVID-19 transmission dynamics. PLOS Digital Health, 2022, 1 (8), pp.e0000098. 10.1371/journal.pdig.0000098. hal-03782635

HAL Id: hal-03782635 https://hal.science/hal-03782635

Submitted on 21 Sep 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Jirsa VK, Petkoski S, Wang H, Woodman M, Fousek J, Betsch C, et al. (2022) Integrating psychosocial variables and societal diversity in epidemic models for predicting COVID-19 transmission dynamics. PLOS Digit Health 1(8): e0000098. https://doi.org/10.1371/journal.pdig.0000098

Editor: Martin G. Frasch, University of Washington, UNITED STATES

Received: July 23, 2021

Accepted: July 28, 2022

Published: August 31, 2022

Copyright: © 2022 Jirsa et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data and codes are available at https://github.com/ HuifangWang/CovidH_TNG.

Funding: This research was supported by European Union's Horizon 2020 research and innovation programme under grant agreement no. 945539 (SGA3) Human Brain Project. Germany's COVID-19 Snapshot Monitoring (COSMO) is funded by the Federal Centre for Health Education (BZgA), the Robert Koch Institute (RKI), the Leibniz **RESEARCH ARTICLE**

Integrating psychosocial variables and societal diversity in epidemic models for predicting COVID-19 transmission dynamics

Viktor K. Jirsa¹*, Spase Petkoski¹*, Huifang Wang¹, Marmaduke Woodman¹, Jan Fousek¹, Cornelia Betsch², Lisa Felgendreff², Robert Bohm^{3,4}, Lau Lilleholt⁴, Ingo Zettler⁴, Sarah Faber⁵, Kelly Shen^{5,6}, Anthony Randal Mcintosh^{5,6}

1 Institut de Neurosciences des Systèmes UMR INSERM 1106, Aix-Marseille Université, 2 University of Erfurt, Erfurt, Germany, 3 Faculty of Psychology, University of Vienna, Vienna, Austria, 4 Department of Psychology and Copenhagen Center for Social Data Science (SODAS) University of Copenhagen, Copenhagen, Denmark, 5 Rotman Research Institute of Baycrest Centre, University of Toronto, Toronto, Canada, 6 Inst Neurosci & Neurotech, Dept of Biomed Physiol and Kinesiol, Simon Fraser University

* viktor.jirsa@univ-amu.fr (VKJ); spase.petkoski@univ-amu.fr (SP)

Abstract

During the current COVID-19 pandemic, governments must make decisions based on a variety of information including estimations of infection spread, health care capacity, economic and psychosocial considerations. The disparate validity of current short-term forecasts of these factors is a major challenge to governments. By causally linking an established epidemiological spread model with dynamically evolving psychosocial variables, using Bayesian inference we estimate the strength and direction of these interactions for German and Danish data of disease spread, human mobility, and psychosocial factors based on the serial cross-sectional COVID-19 Snapshot Monitoring (COSMO; N = 16,981). We demonstrate that the strength of cumulative influence of psychosocial variables on infection rates is of a similar magnitude as the influence of physical distancing. We further show that the efficacy of political interventions to contain the disease strongly depends on societal diversity, in particular group-specific sensitivity to affective risk perception. As a consequence, the model may assist in quantifying the effect and timing of interventions, forecasting future scenarios, and differentiating the impact on diverse groups as a function of their societal organization. Importantly, the careful handling of societal factors, including support to the more vulnerable groups, adds another direct instrument to the battery of political interventions fighting epidemic spread.

Author summary

Societies have a rich inter-individual variability, different cultures, wealth distribution, and ways of functioning, but most epidemic models describe the dynamics of a homogeneous population in different disease stages. The COVID-19 pandemic is exacerbating existing global and national inequalities, which typically hit the most vulnerable groups hardest. The focus of policymakers has been on containing the spread of COVID-19 and mitigating the socioeconomic effects of the pandemic. With the introduction of

Institute for Psychological Information and Documentation (ZPID), and the University of Erfurt. Denmark's COSMO is funded by the Faculty of Social Sciences, University of Copenhagen (to RB and IZ), and Lundbeck Foundation, grant no. R349-2020-592 (to RB).

Competing interests: The authors have declared that no competing interests exist.

psychosocial coupling in epidemic disease models, the consideration of effects linked to societal diversity becomes possible. The model parameters absorb all properties characterizing the diversity of society including psychosocial factors. We estimated the strength and direction of psychosocial causes within a causal inference framework and demonstrated their effect upon the infectious spread to be smaller, but on a similar scale of magnitude as current political interventions and seasonality. Our cause-effect analysis considered minor parametric changes due to psychosocial variation, but has the capacity to integrate a large range of multi-dimensional data including societal, national and cultural differences, political strategies, and trust in the government and media. Given the here estimated size of psychosocial causes, the synergy of these interactions may likely reach scales that cannot be ignored in epidemic spread.

Introduction

The current Coronavirus disease 2019 (COVID-19) pandemic is more than a health crisis; it is also an economic, social and, in many countries, political crisis. Governmental decisions to fight COVID-19 must consider a range of factors including infection spread, health care capacity, as well as economic and psychological variables, all embedded and linked in a worldwide dynamic context. Importantly, these variables influence each other and evolve over time. The specific trajectory of these interactions will determine the consequences for each nation (e.g., deaths, duration of lock-down, economic loss, etc.). Indeed, for the battle against COVID-19 to be successful, it has been suggested to "inform and qualify action with evidence from behavioral and cultural research" [1] to create high acceptance of and compliance with measures, even after restrictions are lifted. While science offers this evidence to inform policy making [2,3], the influence of psychosocial (PS) variables is not integrated into epidemiological models to predict further disease transmission. Feeling risk of contracting the disease but also other, more remote psychosocial constructs such as trust in the regulating bodies and acceptance of policies [1] may be important to take up protective behavior (e.g., physical distancing, wearing a face mask). In current approaches the time invariance of these behaviors is assumed and adapted when an intervention occurs [4]. Any violation of model assumptions reduces the predictive power of the model and limits its validity to short terms. A potential interaction of psychosocial variables with mobility is such a violation: high psychological strain felt during a lock-down may lead people to reject the preventative measures, which is then related to increased mobility [5]. Consequently, model parameters need to be continuously updated to counter for the accumulating model errors [6,7]. As governmental decisions and interventions need to be planned further ahead, the mechanistic correction of the forecast model by integrating the interactions of PS variables with mobility is crucial. Here we use the weekly cross-sectional COVID-19 Snapshot Monitoring (COSMO) data [5,8], which allow rapid and adaptive monitoring of psychosocial variables and gain a real-time insight into the knowledge, perceptions, and behaviors of the population-the "psychological COVID-19 situation" [9]. Our objective is to improve the accuracy of longer-term predictions of epidemic spread models by integrating the influence of PS variables into disease dynamics.

Results

Integrating psychosocial variables in epidemic transmission model of COVID-19

Studies such as COSMO suggest that as PS variables change over time, behavioral responses will change over a similar time scale. The temporal alignment of PS variables, mobility, and

infectious spread provides a framework that enables a causal analysis and testing of this hypothesis [10]. As a first step, we express the influence of PS variables upon behavior as a causal hypothesis, where its coupling strength and direction, i.e. how variables impact eachother, is inferred from the empirical data. Changes in behavior are expressed by modulation of the contact rate and quantified by the mixing matrix β , which describes the number of infection-producing contacts per unit time. R is the reproduction number of an infection and reflects the expected number of cases directly generated by one case in a homogeneously mixed population where all individuals are susceptible to infection [11]. Both quantities, β and R, cannot currently be modified through vaccination or other changes in population susceptibility, but can be modified by physical distancing and other political interventions. These measures, however, are only effective if large parts of the population comply with them. However, certain groups in the population may have less resources and capacity to adapt and thus perceive different levels of affect, e.g. due to age or variations in personal situations, such as a loss of income during a lock-down, exposed working conditions, or restricted living space. We hypothesize that these groups will show different degrees of acceptance of the measures, leading to variation in mobility and ultimately disease transmission, which then again causally loops onto itself, contributing to further changes in PS levels. However, the mixing between the populations is assumed to be homogeneous as a first implementation of this model, and in lack of any data pointing otherwise.

We formalize this emergent and dynamic process in Fig 1, using a compartmental SIR model for infectious disease transmission comprising three populations of susceptible (S) and infected (I) cases, as well as recovered or dead (R) cases. Note that including Exposed group that corresponds to the case with a significant incubation period during which individuals have been infected but are not yet infectious themselves wouldn't have significantly changed the dynamics from the model [12], and moreover this does not seem to be an important feature of the COVID-19 transmission [13]. The total number of people is N = S+I+R. As a government imposes political interventions *J* such as a lock-down, the mixing matrix β and the subsequent reproduction number *R* are reduced. Additional seasonal modulations of β add an



Fig 1. Organization of the PS-SIR model. Psychosocial (PS) variables modulate human mobility through changes of the mixing matrix β_i . As the perceived consequences of political interventions, such as lock-down effects, accrue, the subsequent accumulation changes the affective state ϕ_i of the i-th sub-population. β_i affects ϕ_i through the influence f_i of accumulated PS variables and ϕ_i affects β_i through the influence g_i , e.g., affective risk perception. The index i signals group-specific variation of parameters and variables across the whole population, defining a group structure of vulnerable vs. resilient groups with varying resources and capacity to adapt. Changes in the mixing matrix β_i directly affect the dynamics of the infectious disease in the SIR model.

https://doi.org/10.1371/journal.pdig.0000098.g001

important temporal variation to the epidemic transmission mechanisms [14]. We expand β along two additional dimensions. First, β can slowly evolve over time as a consequence of modulations with the PS variables, which generally evolve on a time-scale of several weeks to months, as also observed in the rate of the changes observed by COSMO data. The time scale separation between infectious transmission and PS variables justifies the mathematical assumption of weak coupling and its expression as an additive modulation [15]. Second, groups within the population can be differentially affected depending on their resources and capacity to adapt to the situation [1], which we separate into *vulnerable* and *resilient* groups for the purposes of the model. The affective state ϕ_i of the *i*-th group is a longer lasting state, which is not caused by a single stimulus but results from an accumulation of experiences. The accumulation of PS effects, u_i, accrues over time due to prolonged restrictions, indicating the capacity of the *i*-th group to adapt to the imposed political interventions. It varies with the pandemic containment efforts such as physical distancing and travel restrictions, but also economic factors such as business closures. Behavioral changes are absorbed in the mixing matrix β_{i} , which establishes the link to the SIR variables and only depends on the political interventions J and the affective state ϕ_i of the *i*-th group.

To infer the strength and direction of the interactions, we use the COSMO data of Germany and Denmark. The measures derived from the COSMO data sets were collected weekly from early-March in Germany (overall N = 11,669) and started a few weeks later in Denmark (overall N = 5,312). The measures focused mainly on the perception of the COVID-19 pandemic, including: AFFECT variables that estimated how serious the threat was to the respondent (e.g., "Is the virus near to you?"; "Is the virus spreading fast?"; "Are you worried about the virus?"), POLICY adjustments and acceptance (e.g., "Should the government close schools?") and TRUST in institutions ("Do you trust the performance of the government?"; "Do you trust the healthcare system?"; see S1 Text in Supplementary Information for more details on questions). Germany and Denmark show both an early increase in the concern around the spread of the virus (AFFECT), and agreement with the governmental interventions (POLICY), especially around school closings. This peaked around March 23, when the country-wide lock-down was instituted in Germany (Fig 2). This trend declined across time, indicating less concern with the pandemic, to levels similar to where they stood when data were first collected. The data from Denmark were collected starting immediately after the country-wide lock-down but showed a comparable declining trend with more recent measurement waves. Fig 2 shows the time-aligned estimates [4] for COVID-19 infection rate and basic reproduction number over the same time period, which were used as a ground truth for comparison with the forward simulations and sampling performed with our model. Fig 2 also shows the mobility of the population based on cell phone tracking indicating sharply reduced mobility during the lockdown and then increasing mobility as AFFECT and POLICY declined [16].

Sampling of psychosocial and behavioral interactions during COVID-19

In order to assess the strength, direction, and significance of the coupling of PS variables on the time-varying reproduction number R, we submit the above extended SIR model with PS interactions to Markov Chain Monte Carlo (MCMC) Bayesian inference. Three parameters are inferred: γ and c are the strength and direction of the couplings g_i and f_i , respectively (Fig 1), and Γ is the time scale of PS accumulation. For simplicity we dropped the group-specific indices of γ and c, as we do not have access to group-specific mobility data from Google. Fig 3 shows the results of the inference and the data used in the process. The two top rows show the time series for the reproduction number R(t) estimates [4] for Germany (DE) and Denmark (DK), respectively, which we are using as ground truth in this workflow, and the



Fig 2. Data available for Germany and Denmark from February 15th to early of May 2020. In the first two rows, estimates for daily infections [4] and reproduction number R are presented as time varying means and standard deviations. Third row shows three selected variables from the AFFECT group in the COSMO surveys (see <u>S1 Text</u> in Supplementary Information for more details; higher values indicate that COVID-19 was seen as something one does not think about all the time, rather not terrifying, and something not to worry about); their weighted mean (derived from PCA) labeled as COSMO(t) is used in the model inversion. The last row presents the scatter plots of the Google mobility report overlaid with lines marking the 7-day weighted moving averages, where the mean of the blue "retail and recreation" variable is used in the model inversion.

https://doi.org/10.1371/journal.pdig.0000098.g002

corresponding fit of the model as lines with the same color scheme. The bottom row shows posterior distributions for model parameters of interest along with the priors (in grey) placed on those parameters. While in Bayesian terms, the coupling strength γ is well identified by the data of both Denmark and Germany, Γ , *c* are identified strongly only for data from Germany. For Denmark these posteriors follow the prior, which are likely due to the lack of COSMO data during the onset of confinement in Denmark. As our primary goal is to determine whether γ is non-zero, i.e., PS factors have an effect on reproduction rate, we implement our null hypothesis as a normal prior on γ , centered at 0.1, i.e., a small effect. MCMC sampling from the posterior distribution allow the data to then identify γ , Γ , *c*. The point estimates (medians) of γ are 0.21 and 0.23 for Germany and Denmark, respectively, and 95% confidence



Fig 3. Inference of PS-SIR model parameters. *Top*: time series are shown for R(t), time varying reproduction rate estimates (see Fig 2), for Germany (DE) and Denmark (DK) in blue and orange circles, respectively, and the fit of the model as lines with the same color scheme. *Middle*: time series are shown for COSMO AFFECT data, for Germany and Denmark in blue and orange circles, respectively, and the fit of the model as lines with the same color scheme. Lower values indicate higher negative affect such as more fear and worries due to the Coronavirus. *Bottom*: MCMC posterior distributions for key model parameters (γ : coupling strength of g_{ij} , Γ : time scale of psychosocial accumulation; *c*: coupling strength of f_i) for Germany and Denmark (blue and orange histograms, respectively) and their respective priors (grey lines). On the left, the histograms of posterior samples for γ for both countries, which deviate significantly from the prior (grey line); the data thus identify a strong and robust effect of PS factors on SIR dynamics. Similar effects are seen for Γ and *c* for Germany, while Denmark follows the prior on these parameters likely because of missing COSMO data at the onset of confinement.

https://doi.org/10.1371/journal.pdig.0000098.g003

intervals are (0.17, 0.25) and (0.18, 0.30) for Germany and Denmark, respectively. This indicates that PS variables significantly impact the reproduction rate with a non-zero and positive coupling strength, i.e. the less stress people felt about COVID-19, the higher *R*. This defines a baseline, which allows individual parametrization of the models for Germany and Denmark.

Note that the surrogate model corresponding to a null hypothesis where $\gamma = 0$ is available to the inference process, because the prior on γ covers the value 0 with higher probability than the posterior values that we report. This indicates that the posterior has been well informed by the data. We could have evaluated an equivalent model with γ fixed to 0, and performed a model comparison, but this would be identical to allow the model to explore the parameter space around $\gamma = 0$ which is what happens with MCMC with the prior we have chosen.

Exploring the effects of PS variation on pandemic spread

We initialize each model with its respective past values for epidemic, mobility, and PS variables and provide simulations for the upcoming months. To estimate robustness of the forecast

simulations and sensitivity upon the inferred parameters, we run simulations for all values of coupling parameters sampled from the posterior distributions. The panels of Fig 4 show a comparison of simulations for the scenarios with and without PS coupling, displaying the fore-cast trajectory evolution until April 2021 for the fraction of susceptible *S* and daily new infections derived from the infected individuals *I*, reproduction number *R*, affective state ϕ_i , and estimated strength of political interventions *J*, which for March of 2020 is set so that the dynamics after the intervention would follow the observed infections and *R*. For the subsequent governments' relaxations and lockdowns, *J* is set to follow the relative strictness between the measures at different periods.

The left column of Fig 4 displays the situation for Germany, the right for Denmark. In addition to the good agreement with the estimates for the pandemic spread until the summer 2020 [4], the model is also capturing the subsequent estimated and reported peaks and declines in the infection rates of both countries all the way to the spring of 2021 [17,18]. Further differentiation comprises group structure as indexed by *i*, simulating the population into groups of vulnerable and resilient based on different levels of PS variables in response to political interventions. The group differentiation is realized by assigning different values to coupling parameters c_i and Γ_i .

The seasonality of transmission [14] is included in the simulations where more transmissions are assumed in autumn/winter than in spring/summer. Together, PS variables and seasonality cause the predicted epidemic wave in autumn/winter of 2020 to be larger than the previous one in spring 2020, despite the same stringency level of lock-down. This effect is also partly responsible (together with the government interventions) for the lower level of infections observed during May and June 2020 than later on in autumn. For every simulation of the SIR model without PS variables (magenta dashed lines in Fig 4A), the efficacy of political interventions *J* is overestimated, showing an order of magnitude smaller second wave, than the one that was observed and predicted with the full model. When comparing the forecast epidemic spread with and without PS coupling ($\gamma = 0$), the political intervention I would have proven sufficient to control the epidemic spread in the latter case, but not the former, because the PS variables cause a surge of infections towards the second part of 2020. This is even better illustrated when comparing the observed [17,18] curves for S and new daily infections in S5 Fig, which contain the same peaks and values within the same ranges as the model, up to the May 2021, i.e. S between 85–90%. The model without the PS subsystem misses the subsequent peaks, and has an order of magnitude smaller ratio of susceptible (S < 2%). Even though the mean inferred impact of the PS subsystem is slightly lower for Germany than for Denmark, values for R and the second infection wave related to the seasonality is predicted to be larger in Germany.

The sensitivity of the simulations to the choice of parameters is explored in Fig 4B. Here the coupling parameters γ are drawn from the posterior distributions for a sample of 4,000 model realizations for each country, and then used in the simulations with all other parameter settings equal. The shaded areas indicate a confidence interval of 95%. All simulated trajectories show qualitatively the same behavior, that is a surge of infection towards the end of 2020. To provide a quantitative impression of the strength of PS coupling as compared to lock-down and seasonal variation, Fig 4C shows a comparison of the impact of different interventions with the seasonality and the PS influence, illustrated for the case of Denmark. Analysis for Germany provides the same results and is provided in the Supplementary Information S2 Fig. The inclusion of group structure into the PS coupling generally enhances its strength, where more heterogeneity causes stronger PS effects (see also Supplementary Information S1 Fig for further elaboration). The impact is computed as the difference of cumulative infections with and without cause (intervention, seasonality, and PS variables). Given the cumulative nature of PS



Fig 4. Forecast of the evolution of COVID 19 spread for Germany and Denmark. (A) Simulations of the SIR (dashed magenta), PS-SIR model with (black) and without group structure (dashed blue) for Germany (left) and Denmark (right) as well as the PS-SIR model split up for populations assuming 50% vulnerable (red) or resilient (green) citizens, simulated until May 2021. "S" denotes the ratio of susceptible, new infections are derived from "I", "R" is the reproduction factor, phi perceived negative AFFECT, and J political interventions. The default parameters in all the plots are from a run of the chain that best fits the data shown in Fig 2 up to mid-May 2020. Simulations contain all the government interventions up to the spring of 2021. (B) Ensemble projection of infections for all the combinations of inferred parameters for both countries (light red and blue lines), the mean (red and blue line) and the areas of 95% confidence (shaded red and blue). (C) Estimated strengths of interventions on infectious spread for Denmark (results for Germany are comparable and provided in the Supplementary Information, S2 Fig): Red is the impact of the government interventions (the largest on 19/03, and the subsequent relaxations on 13/05 and 30/06); blue is seasonality (the maximum impact, and the impact for 2 months after the relaxation on 14/04 and before the predicted intervention in autumn; and dark green is the predicted impact in autumn 2020, separated per group.

https://doi.org/10.1371/journal.pdig.0000098.g004

effects, the effect strength has been inferred for a particular time after the onset of the cause, in particular the cumulative impact of PS is shown before the relaxation on 14/04 and before the predicted intervention in autumn (light green); furthermore, the cumulative impact of group structure is also shown (dark green), clearly demonstrating its epidemic spread-reinforcing effect. Political interventions via mobility restrictions show the strongest effect, seasonality is factor 2 smaller than mobility constraint, and a factor 3 to 4 larger than PS factors.

Methods and model

COVID-PS Model

We express the influence of ϕ_i and β_i of the *i*-th group by the coupling functions $f_i(u_i, J, \phi_i)$ and $g_i(\phi_i, J, \beta_i)$, which shall be inferred from the data. As a first step in model building, we recognize similarities with many physical and biological systems, which are coupled indirectly through a slowly evolving dynamic medium. Such slow modulations of a coupling medium are effectively equivalent to a linear weak coupling of the dynamic systems, prescribing their interaction around a working point as additive in a first approximation, but leaving the mathematical form of the actual interaction functions themselves open. We harness this effect to extend the transmission dynamics of the SIR model to include psychosocial interactions and infer the precise form of the influence functions using Monte Carlo Markov Chain (MCMC) methods. The full set of equations reads:

$$\dot{S}_i = \Lambda_i - \beta_i S_i \sum_{j=l}^{ll} I_j - \mu S_i \tag{1}$$

$$\dot{I}_i = \beta_i S_i \sum_{j=I}^{II} I_j - \alpha I_i - \mu I_i$$
⁽²⁾

$$\dot{\boldsymbol{\beta}}_i = -(\boldsymbol{\beta}_i - \boldsymbol{\beta}_0) - \boldsymbol{g}_i(\boldsymbol{\phi}_i, \boldsymbol{J}, \boldsymbol{\beta}_i)$$
(3)

$$\dot{\phi}_i = -(\phi_i - \phi_0) - f_i(J, \beta_i, t) \tag{4}$$

where the coupling functions $g_i(\phi_i, J, \beta_i)$ and $f_i(J, \beta_i)$ t) affect the growth rates of the mixing matrix β_i , effective state ϕ_i , while the assumed infectious time is set at 14 days, i.e. $\alpha = 1/14$ [4], instead of being modelled as a distribution [12]. The time is hence implicit and set in days. The latent period of the disease, even though important for the evolution on shorter time-scales [19], is ignored in our model due to the focus on the long term evolution. For similar reasons and to focus on the impact of the PS variables, the model also does not contain removed population, and S and I are relative to the total population, so that S+I = 1. Since we are interested in the dynamics of the pandemic variables, all the other units are assumed phenomenological and set accordingly to this.

The death rate is set to be equal for the both groups, $\mu = 1/365/80$ and the total birth rate of both groups the same, with each group having $\Lambda_i = \mu/2$. These rates are at much slower timescales than the relevant ones for the diseases, but we have nevertheless introduced them to allow for a consistent division on groups, which is performed through Λ_i . For simplicity, and due to lack of other priors, the mixing matrix is assumed to be identical among the groups.

The back-coupling of β_i on ϕ_i is communicated via an accumulation of lock-down effects, aka deviations from normal mobility, through an auxiliary variable u_i . For both types of interactions, the political intervention J is included in the coupling and subject to the inference. The coupling function f_i is assumed to increase linearly and then saturates at a maximal value of 1, whereas g_i shows the same behavior but remains restricted to the linear signal range. The strength and direction of the coupling is equivalent to the scale and sign of γ_i and c_i of the

couplings g_i and f_i , respectively, subject to inference from empirical data using a stochastic model (see below). As we will not discuss group-specific mobility changes due to changes in PS variables, we drop the index of γ_i , leading to $g_i(\phi_i, J, \beta_i) = J\beta_i + \gamma_i \cdot (\phi_i - \phi_0)$ and $f_i(J, \beta_i, t) = hJ + c_i \tanh(2u_i)$, so that the following closed loop equations appear:

$$\hat{\beta}_i = -(\beta_i - \beta_0) - J\beta_i - \gamma_i \cdot (\phi_i - \phi_0)$$
(5)

$$\dot{\phi}_i = -(\phi_i - \phi_0 + hJ) - c_i \tanh(2u_i) \tag{6}$$

$$\dot{u}_i = -\Gamma_i / \sigma(\beta_i - \beta_0) \tag{7}$$

Here *u* is an auxiliary variable for the accumulation of restrictions-related fatigue. As a such, $u_i = 0$ for J = 0, i.e. when there are no restrictions, and during the relaxation of restrictions u_i immediately decreases such that $u_i(t) = \tanh^{-1} \left(\tanh(u_i(t-1)) \left(\frac{J(t)}{J(t-1)} \right)^2 \right)$, which

comes from the condition $u_i(t) = u_i(t-1) \left(\frac{J(t)}{J(t-1)}\right)^2$. During restrictions, it linearly increases with their severity, which is accounted through the impact on the mixing matrix, i.e. larger the deviation of β_i from β_0 , larger is the fatigue. In the same time, each group can be differently affected by the interventions, and this is scaled with the parameter Γ_i . However, the way this affects the affective state (or the happiness) ϕ , is set to be a sigmoid, saturating at levels that are driven by c_i , which is inferred from the data together with Γ_i and γ_i . Here $\Gamma_i/\sigma <<1$ reflects the time scale separation, that is both mobility via β_i and affective state via ϕ_i evolve on a time scale of the order of 1, whereas PS effects accumulate on a time scale of the order of Γ_i , hence we set $\sigma = 30$. The impact of the government interventions J on ϕ is smaller than on β , and hence h = 0.2, meaning the mixing matrix is stronger influenced by the interventions, than is the affective state ϕ . The overall coupling of the SIR and PS sub-systems is driven by the parameter γ_i , which describes how much PS variables impact the dynamics of the epidemic.

For the asymmetry between the group parameters c_i and Γ_i , whose mean values were inferred, we set the first group to have 9 times larger values than the second, corresponding to 80% larger or smaller values than the mean for each respective group.

It should be noted that depending on the nature of the phenomena, the coupling between them is chosen to be sigmoid, if the impact is supposed to saturate, or linear otherwise. These are the simplest choices due to the phenomenological nature of the model, even though the coupling functions [20] and their proper recovery [21] are of crucial importance for biological oscillations [22].

To describe the logic behind the equations, absence of interventions or an impact of the PS variables, the first term in $\dot{\beta}$ leads to a relaxation of the transmission rate towards pre pandemic levels over time. The second term lowers this asymptotic value due to government interventions, and the last term increases it due to the decrease in the affective state ϕ_i , as compared to its base value $\phi_0 = 1$. The strength of this effect is driven by the coupling variable γ_i that is being inferred. The affective state ϕ_i is similarly relaxing to its base value ϕ_0 in absence of interventions *J*, which make it to decrease exponentially to a new level dependent on the strength of *J*. Similar impact on ϕ_i is observed by the accumulated fatigue, which is a sigmoid with a time constant of months, and with a level set by the inferred parameters c_i .

To include the impact of seasonality on the transmission ability of the virus [14], we adapt sinusoidal modulation of β_0 with a period of one year and the peak being in Mid-January for

the northern hemisphere. The seasonally varying $\beta_0(t)$ hence reads:

$$\beta_0(t) = \beta_0 - \frac{J_s}{2} \left(1 - \cos \frac{2\pi (t - t_0)}{365}\right)$$

with J_s being the magnitude of the variation that depends on the latitude and the climate. For both Germany and Denmark, it is fixed at 40%, which is the level predicted for New York [14].

To calculate the relative impact of the different parts of the model (PS part, seasonality, and the government interventions), for each country we run forward simulations that exclude certain parts of the model and we calculate the difference in the observed numbers for given periods of time.

The model is numerically integrated in Matlab using the Heun integration method with a time step of 0.05 days. A moving average of 1 day was then used to bring the results of the model to the same resolution as the empirical data. R(t) is calculated as $R(t) = \frac{\Lambda_i \beta_i}{\mu(\alpha+\mu)}$, and for better comparison between groups and models it is always normalized as R of different models/groups were referring to a same size of population (effectively the values in the case of groups are multiplied by 2).

The statistical model was implemented in v2.22.1 of the Stan probabilistic programming language [23], with parameters sampled using the No-U-Turn variant [24] of Hamiltonian Monte Carlo [25]. The posterior draws were verified with the Stan 'diagnose'utility, which checks for transitions that hit the maximum tree depth, divergent transitions, low estimated Bayesian fraction of missing information (E-BFMI), low effective sample sizes and \hat{R} values below 1.1 (indicating convergence of the MCMC method). Posterior distributions for key parameters and posterior predictive distributions for data were constructed, verifying that the data lie in a 95% confidence interval, to ensure model fit is reasonable.

Grouping PS variables in the COSMO data

Given the importance of the group structure on the relation between PS variables and SIR suggested by the model, we further analyzed the COSMO survey data to determine if significant grouping variables could be extracted. We tested age (young, middle, and old age), sex, and education (three levels) and compared the aggregate PS variables across all available waves using multivariate Partial Least Squares (PLS) analysis [26,27]. Employment status and household income were less consistently collected so were not considered for this analysis. In both German and Danish data, differences based on sex (Fig 5) and age (Supplementary Information S1 Fig) were observed. Differences related to education were less reliable but existent. We also observed that the aggregate PS measure was reliably related to subjective psychological stress and resilience in a subset of the COSMO data, suggesting the aggregate captured broader psychosocial aspects beyond risk perception (see Supplementary Information S4 Fig). For Germany, data collection started earlier and shows an initial overlapping transient with a faster rebound for males than for females (Fig 5 left), resulting in a group difference which remains preserved as time evolves. The same situation is observed for Denmark, even more pronounced (Fig 5 right), but without the initial transient as data collection started later. The interaction of age and sex was not significant, indicating these were additive factors in these data. Older adults and females showed larger overall values on the PS effect, suggesting these groups felt higher affective risk and that the COVID-19 pandemic was more of a crisis relative to younger groups or males.



Fig 5. Sex differences in PS indicators. The dominant effect (both p < 0.001 from permutation test) of a multivariate Partial Least Squares analysis of mean change over time of PS indicators in COSMO data from Germany (A-B) and Denmark (C-D) by sex. AFFECT (AFF), POLICY (POL) and TRUST variables were included if data were available from all waves in each of the COSMO datasets. See Table A in <u>S1 Text</u> in Supplementary Information for a complete list of variables. Error bars: Bootstrap estimated 95% CI. Bootstrap ratios are the singular value weights divided by their standard error and are roughly equivalent to a z-score. The shape of the Psychosocial score curve relative to the PS Indicators suggests an increase in affect risk perception and great institutional and policy trust in March, followed by a general relative reduction.

https://doi.org/10.1371/journal.pdig.0000098.g005

Discussion

Although mechanisms of COVID-19 transmission are indiscriminate, this may not extend to a complex society fighting epidemic spread [28]. Societies have a rich inter-individual variability, different cultures, wealth distribution, and ways of functioning, but most epidemic models describe the dynamics of a homogeneous population in different disease stages. The model parameters absorb all properties characterizing the diversity of society including psychosocial factors. With the introduction of psychosocial coupling in epidemic disease models, the consideration of effects linked to societal diversity becomes possible.

We estimated the strength and direction of psychosocial causes within a causal inference framework and demonstrated their effect upon the infectious spread to be smaller, but on a similar scale of magnitude as current political interventions (approx. factor 8 smaller than lock-down effects and social distancing) and seasonal variation (approx. factor 4-5 smaller). In particular, affect variables that estimated how serious the threat was to the respondent provide strong evidence for the causal effect of psychosocial factors on mobility and subsequently infectious spread. Our cause-effect analysis considered minor parametric changes due to psychosocial variation, but has the capacity to integrate a large range of multi-dimensional data including societal, national and cultural differences, political strategies, and trust in the government and media. For instance, multidimensional approaches could integrate additional psychological and sociocultural indicators, affecting the mixing matrix, thereby establishing a battery of mutually dependent behaviors. Given the here estimated size of psychosocial causes, the synergy of these interactions may likely reach scales that cannot be ignored in epidemic spread. Furthermore, heterogeneity within a society generates groups that are not equally affected by the same political intervention. The group structuring differences are plotted in Fig 5 and supplemental figures, demonstrating that a highly heterogeneous society would show acceleration and augmentation of infections across the entire population due to group structure (S1 Fig). Despite the fact that the resilient population experiences no change in its affective state, the uniform mixing of the populations causes increases of total numbers of infections in both groups, vulnerable and resilient. A detailed quantitative understanding of these effects provides an added means to guide political decisions to influence the infectious spread by considering parameters linked to demographic factors of societal organization. In the continuum spanned by group heterogeneity and timing of interventions, the choice of the best lock-down duration will depend on the heterogeneity within the social structure and allow the optimization of political interventions for a given society, and thus minimize economic damage while respecting the efficiency for epidemic transmission control. This is particularly important as interventions have optimal working domains (such as lock-down strength or duration), but outside of these windows they are inefficient. For example, in S1 Fig A for Denmark, we show that extending lock-down durations do not alter the trajectory of epidemic spread significantly. Alternatively, the same mechanism in our model may be used to increase the efficacy of political interventions by changing societal factors, in particular reducing societal heterogeneity and increasing support to the more vulnerable groups [1].

Partly because of a lack of specific data in the COSMO surveys, we have left explicit economic factors aside in this discussion, although such factors would be relevant to the grouping structure. It has been suggested that psych-socio-economic variables such as income, education, and ethnicity play an important role in the transmission of infectious diseases [29–31], not only because they reflect resilience, but also because they influence the adoption of protective behaviors during pandemics [32]. As a consequence, models taking into consideration PS-SIR interactions can provide stronger support for political interventions that support less resilient groups socially and economically as this is a factor in reducing heterogeneity [33,34]. The model framework also underscores that psychological knowledge from personality and health psychology can be put to very concrete use in a crisis [35]. Future efforts should quantitatively assess the impact of communication and behavioural interventions that address PS factors to enhance compliance. An important caveat in our approach is the uncertainty of the ground truth, against which we are validating our results. Underreporting of the cases has varied greatly during the course of the pandemics [36], but still several reliable datasets appeared trying to infer and fit the pandemics evolution as reliably as possible [37]. Using this as ground truth, we demonstrate that the model with PS variables can qualitatively capture the evolutions of infections, which is not the case when these are omitted.

Similarly, besides the mobility which is the focus here, reactive behavioural changes, in particular, how we come into contact with one another, are also impacting the mixing matrix [38]. It would be important for these to be also accounted for in future models that account the PS factors. In the same direction, having a true mixing matrix β_{ij} with group-specific transmission rate modulators, instead of the assumption about equal risk of infections per group, would also make the model closer to reality. Accounting for the latent period of the disease is also important for the future modifications, especially since it is strongly influencing the growth rate of the incidence [19]. In the same line, the impact of the seasonality in the spread of COVID-19 pandemic seems to be well established [39,40], and in further developments of the model, its impact would need to be also inferred from the multiseasonal data. Finally, COVID-19 pandemic is continuously evolving, with new variants appearing with different virulence, but also different vaccination strategies across countries [41], which both would need to be included in future models.

The COVID-19 pandemic is exacerbating existing global and national inequalities, which typically hit the most vulnerable groups hardest [42,43]. The focus of policymakers has been on containing the spread of COVID-19 and mitigating the socioeconomic effects of the pandemic. Our findings demonstrate that addressing these vulnerabilities nation- or even worldwide through policy action (for instance, temporary basic income strategy [28,44,45] may not only be a mean to fight the socioeconomic consequences of COVID-19, but also represent a mean to directly fight its spread across all groups, both vulnerable and resilient, through non-pharmaceutical actions containing the disease.

Supporting information

S1 Text. Table A. COSMO PS and psychological well-being indicators included for analysis.

(DOCX)

S1 Fig. Covid-19 spread and cause-effect scenarios under variation of lock-down duration, PS coupling strength and societal heterogeneity. Parameter sweeps for (A) the government interventions, (B) coupling strength of the PS system, and (C) the group specific parameters. The results for the default parameters are always green. For clarity, the impact of seasonality is not included and only the current interventions are used to better identify the impact of the government interventions and the PS subsystem and its heterogeneity. All simulations are performed for Denmark.

(TIFF)

S2 Fig. Estimated strengths of Interventions on infectious spread for Germany (same as in Fig 4C).

(TIFF)

S3 Fig. Relationship between subjective psychological well-being and resilience and PS indicators. First latent variables (both p<0.001) of a multivariate PLS analysis relating psychological well-being (PSY/RES) to the AFFECT aggregate in German (A-B) and Danish (C-D) COSMO data. Pearson correlation coefficients (A and C) show a stable relationship across

time. See Table A in <u>S1 Text</u> for a complete list of variables. Bootstrap ratios are singular value weights divided by their standard error and are roughly equivalent to a z-score. (TIFF)

S4 Fig. Age differences in PS indicators. The dominant latent variable (both p<0.001 by permutation test) of a multivariate PLS analysis of mean change over time of PS indicators in German (A-B) and Danish (C-D) COSMO data by age group. Subjects were grouped into three equally sized bins of age categories. In the German data, age categories were 18–37, 38–55 and 56–87 years. In Danish data, age categories were 18–49, 50–63 and 64–92 years. AFFECT (AFF), POLICY (POL) and TRUST variables were included if data were available from all waves in each of the COSMO datasets. See Table A in <u>S1 Text</u> for a complete list of variables. Error bars: 95% CI.

(TIFF)

S5 Fig. Data available for Germany and Denmark from February 15th to early of May 2021. Estimates and reported daily cases and reproduction number R are presented as time varying means and standard deviations. Data recovered from https://covid19.healthdata.org/. (TIFF)

Acknowledgments

Germany's COVID-19 Snapshot Monitoring (COSMO) is a joint project of the University of Erfurt (Cornelia Betsch [PI], Lars Korn, Philipp Sprengholz, Philipp Schmid, Lisa Felgendreff, Sarah Eitze), the Robert Koch Institute (RKI; Lothar H. Wieler, Patrick Schmich), the Federal Centre for Health Education (BZgA; Heidrun Thaiss, Freia De Bock), the Leibniz Centre for Psychological Information and Documentation (ZPID; Michael Bosnjak), the Science Media Center (SMC; Volker Stollorz), the Bernhard Nocht Institute for Tropical Medicine (BNITM; Michael Ramharter), and the Yale Institute for Global Health (Saad Omer).

Author Contributions

Conceptualization: Viktor K. Jirsa, Spase Petkoski, Anthony Randal Mcintosh.

- Data curation: Viktor K. Jirsa, Jan Fousek, Cornelia Betsch, Lisa Felgendreff, Robert Bohm, Lau Lilleholt, Ingo Zettler, Sarah Faber.
- **Formal analysis:** Viktor K. Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Kelly Shen, Anthony Randal Mcintosh.

Funding acquisition: Viktor K. Jirsa, Cornelia Betsch, Robert Bohm, Ingo Zettler.

- **Investigation:** Viktor K. Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Jan Fousek, Cornelia Betsch, Lisa Felgendreff, Robert Bohm, Lau Lilleholt, Ingo Zettler, Sarah Faber, Kelly Shen, Anthony Randal Mcintosh.
- **Methodology:** Viktor K. Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Jan Fousek, Cornelia Betsch, Lisa Felgendreff, Robert Bohm, Lau Lilleholt, Ingo Zettler, Sarah Faber, Kelly Shen, Anthony Randal Mcintosh.
- Project administration: Viktor K. Jirsa.
- **Resources:** Viktor K. Jirsa, Cornelia Betsch, Robert Bohm, Ingo Zettler, Anthony Randal Mcintosh.

Software: Viktor K. Jirsa, Spase Petkoski, Marmaduke Woodman, Jan Fousek.

Supervision: Viktor K. Jirsa.

Validation: Viktor K. Jirsa, Spase Petkoski.

Visualization: Viktor K. Jirsa, Spase Petkoski, Marmaduke Woodman, Jan Fousek.

- Writing original draft: Viktor K. Jirsa, Spase Petkoski, Huifang Wang, Marmaduke Woodman, Jan Fousek, Cornelia Betsch, Lisa Felgendreff, Robert Bohm, Lau Lilleholt, Ingo Zettler, Sarah Faber, Kelly Shen, Anthony Randal Mcintosh.
- Writing review & editing: Viktor K. Jirsa, Spase Petkoski, Marmaduke Woodman, Jan Fousek.

References

- Habersaat KB, Betsch C, Danchin M, Sunstein CR, Böhm R, Falk A, et al. Ten considerations for effectively managing the COVID-19 transition. Nat Hum Behav. 2020 Jul; 4(7):677–87. https://doi.org/10.1038/s41562-020-0906-x PMID: 32581299
- 2. Betsch C. How behavioural science data helps mitigate the COVID-19 crisis. Nat Hum Behav. 2020 May; 4(5):438–438. https://doi.org/10.1038/s41562-020-0866-1 PMID: 32221514
- Metcalf CJE, Morris DH, Park SW. Mathematical models to guide pandemic response. Scinece. 2020; 369(6502):368–9. https://doi.org/10.1126/science.abd1668 PMID: 32703861
- Flaxman S, Mishra S, Gandy A, Unwin HJT, Mellan TA, Coupland H, et al. Estimating the effects of non-pharmaceutical interventions on COVID-19 in Europe. Nature. 2020 Jun; <u>https://doi.org/10.1038/</u> s41586-020-2405-7 PMID: 32512579
- Betsch, Cornelia; Wieler, Lothar; Bosnjak, Michael; Ramharter, Michael; Stollorz, Volker; Omer, Saad; Korn, Lars; Sprengholz, Philipp; Felgendreff, Lisa; Eitze, Sarah; Schmid P. German COVID-19 Snapshot MOnitoring (COSMO Germany). PsychArchives. 2020;
- 6. Siegenfeld AF, Taleb NN, Bar-Yam Y. What models can and cannot tell us about COVID-19. Proceedings of the National Academy of Sciences of the United States of America. 2020. p. 16092–5.
- Bertozzi AL, Franco E, Mohler G, Short MB, Sledge D. The challenges of modeling and forecasting the spread of COVID-19. PNAS. 2020 Jul;202006520. https://doi.org/10.1073/pnas.2006520117 PMID: 32616574
- 8. Böhm R, Lilleholt L, Zettler I, COSMO Denmark group. Denmark COVID-19 Snapshot MOnitoring (COSMO Denmark): Monitoring knowledge, risk perceptions, preventive behaviours, and public trust in the current coronavirus outbreak in Denmark. PsychArchives. 2020;
- Betsch C, Wieler LH, Habersaat K. Monitoring behavioural insights related to COVID-19. Lancet. 2020 Apr; 395(10232):1255–6. https://doi.org/10.1016/S0140-6736(20)30729-7 PMID: 32247323
- 10. Pearl J. Causal inference in statistics: An overview. Stat Surv. 2009; 3:96–146.
- 11. Fine P, Eames K, Heymann DL. "Herd Immunity": A Rough Guide. Clin Infect Dis. 2011 Apr; 52(7):911– 6. https://doi.org/10.1093/cid/cir007 PMID: 21427399
- 12. Choisy M, Gugan J-F, Rohani P. Mathematical Modeling of Infectious Diseases Dynamics. Encycl Infect Dis. 2007;379–404.
- He X, Lau EHY, Wu P, Deng X, Wang J, Hao X, et al. Temporal dynamics in viral shedding and transmissibility of COVID-19. Nat Med. 2020; 26(5):672–5. <u>https://doi.org/10.1038/s41591-020-0869-5</u> PMID: 32296168
- Kissler SM, Tedijanto C, Goldstein E, Grad YH, Lipsitch M. Projecting the transmission dynamics of SARS-CoV-2 through the postpandemic period. Science (80-). 2020 May; 368(6493):860–8. https://doi. org/10.1126/science.abb5793 PMID: 32291278
- Rubin JJ, Rubin JE, Ermentrout GB. Analysis of Synchronization in a Slowly Changing Environment: How Slow Coupling Becomes Fast Weak Coupling. Phys Rev Lett. 2013 May; 110(20):204101. https:// doi.org/10.1103/PhysRevLett.110.204101 PMID: 25167415
- 16. COVID-19 Community Mobility Reports. https://www.google.com/covid19/mobility/. 2020.
- 17. https://covid19.healthdata.org/germany?view=infections-testing&tab=trend&test=infections.
- 18. https://covid19.healthdata.org/denmark?view=infections-testing&tab=trend&test=infections.
- Heng K, Althaus CL. The approximately universal shapes of epidemic curves in the Susceptible-Exposed-Infectious-Recovered (SEIR) model. Sci Rep. 2020; 10(1). https://doi.org/10.1038/s41598-020-76563-8 PMID: 33168932

- Stankovski T, Pereira T, Mcclintock PVE, Stefanovska A. Coupling functions: Universal insights into dynamical interaction mechanisms. Rev Mod Phys [Internet]. 2017; 89(October-December):1–50. Available from: https://arxiv.org/pdf/1706.01810.pdf
- Stankovski T, Duggento A, McClintock PVE, Stefanovska A. Inference of time-evolving coupled dynamical systems in the presence of noise. Phys Rev Lett. 2012; 109(2):1–5. <u>https://doi.org/10.1103/</u> PhysRevLett.109.024101 PMID: 23030162
- 22. Stankovski T, Petkoski S, Smith AF, McClintock PVEE, Stefanovska A, Raeder J, et al. Alterations in the coupling functions between cortical and cardio-respiratory oscillations due to anaesthesia with propofol and sevoflurane. Philos Trans R Soc A Math Phys Eng Sci [Internet]. 2016; 374(2067):20150186. Available from: http://rsta.royalsocietypublishing.org/content/374/2067/20150186 https://doi.org/10. 1098/rsta.2015.0186 PMID: 27045000
- Carpenter B, Gelman A, Hoffman MD, Lee D, Goodrich B, Betancourt M, et al. Stan: A Probabilistic Programming Language. J Stat Softw. 2017; 76(1).
- 24. Hoffman MD, Gelman A. The no-U-turn sampler: Adaptively setting path lengths in Hamiltonian Monte Carlo. J Mach Learn Res. 2014;
- 25. Betancourt M. A Conceptual Introduction to Hamiltonian Monte Carlo. 2017 Jan;
- McIntosh AR, Lobaugh NJ. Partial least squares analysis of neuroimaging data: applications and advances. Neuroimage. 2004 Jan; 23:S250–63. https://doi.org/10.1016/j.neuroimage.2004.07.020 PMID: 15501095
- Krishnan A, Williams LJ, McIntosh AR, Abdi H. Partial Least Squares (PLS) methods for neuroimaging: A tutorial and review. Neuroimage. 2011 May; 56(2):455–75. <u>https://doi.org/10.1016/j.neuroimage</u>. 2010.07.034 PMID: 20656037
- Molina GG, Eduardo O-J. Temporary Basic Income: Protecting Poor and Vulnerable People in Developing Countries. United Nations Dev Program. 2020;
- 29. Braveman P. Accumulating Knowledge on the Social Determinants of Health and Infectious Disease. Public Health Rep. 2011 Sep; 126(3_suppl):28–30. <u>https://doi.org/10.1177/00333549111260S306</u> PMID: 21836734
- Bambra C, Riordan R, Ford J, Matthews F. The COVID-19 pandemic and health inequalities. J Epidemiol Community Health. 2020 Jun; jech-2020-214401. <u>https://doi.org/10.1136/jech-2020-214401</u> PMID: 32535550
- Bonds MH, Keenan DC, Rohani P, Sachs JD. Poverty trap formed by the ecology of infectious diseases. Proc R Soc B Biol Sci. 2010 Apr; 277(1685):1185–92. https://doi.org/10.1098/rspb.2009.1778 PMID: 20007179
- Bish A, Michie S. Demographic and attitudinal determinants of protective behaviours during a pandemic: A review. Br J Health Psychol. 2010 Nov; 15(4):797–824. <u>https://doi.org/10.1348/</u> 135910710X485826 PMID: 20109274
- Kumar S, Piper K, Galloway DD, Hadler JL, Grefenstette JJ. Is population structure sufficient to generate area-level inequalities in influenza rates? An examination using agent-based models. BMC Public Health. 2015 Dec; 15(1):947. https://doi.org/10.1186/s12889-015-2284-2 PMID: 26400564
- 34. Walensky RP, del Rio C. From Mitigation to Containment of the COVID-19 Pandemic. JAMA. 2020 May; 323(19):1889.
- Michels M, Glöckner A, Giersch D. Personality Psychology in Times of Crisis: Profile-specific Recommendations on how to deal with COVID-19. PsyArXiv Prepr. 2020;
- Lau H, Khosrawipour T, Kocbach P, Ichii H, Bania J, Khosrawipour V. Evaluating the massive underreporting and undertesting of COVID-19 cases in multiple global epicenters. Pulmonology. 2021; 27(2). https://doi.org/10.1016/j.pulmoe.2020.05.015 PMID: 32540223
- Barber RM, Sorensen RJD, Pigott DM, Bisignano C, Carter A, Amlag JO, et al. Estimating global, regional, and national daily and cumulative infections with SARS-CoV-2 through Nov 14, 2021: a statistical analysis. Lancet. 2022;1–30.
- Prem K, van Zandvoort K, Klepac P, Eggo RM, Davies NG, Cook AR, et al. Projecting contact matrices in 177 geographical regions: An update and comparison with empirical data for the COVID-19 era. PLoS Computational Biology. 2021. https://doi.org/10.1371/journal.pcbi.1009098 PMID: 34310590
- Liu X, Huang J, Li C, Zhao Y, Wang D, Huang Z, et al. The role of seasonality in the spread of COVID-19 pandemic. Environ Res. 2021; 195. https://doi.org/10.1016/j.envres.2021.110874 PMID: 33610582
- Yin C, Zhao W, Pereira P. Meteorological factors' effects on COVID-19 show seasonality and spatiality in Brazil. Environ Res. 2022; 208. https://doi.org/10.1016/j.envres.2022.112690 PMID: 34999027
- Vasireddy D, Vanaparthy R, Mohan G, Malayala SV, Atluri P. Review of COVID-19 Variants and COVID-19 Vaccine Efficacy: What the Clinician Should Know? J Clin Med Res. 2021; 13(6). <u>https://doi.org/10.14740/jocmr4518</u> PMID: 34267839

- Xafis V. 'What is Inconvenient for You is Life-saving for Me': How Health Inequities are playing out during the COVID-19 Pandemic. Asian Bioeth Rev. 2020 Jun; 12(2):223–34. <u>https://doi.org/10.1007/s41649-020-00119-1</u> PMID: 32427219
- Okonkwo NE, Aguwa UT, Jang M, Barré IA, Page KR, Sullivan PS, et al. COVID-19 and the US response: accelerating health inequities. BMJ Evidence-Based Med. 2020 Jun;bmjebm-2020-111426. https://doi.org/10.1136/bmjebm-2020-111426 PMID: 32493833
- 44. Alberti PM, Lantz PM, Wilkins CH. Equitable Pandemic Preparedness and Rapid Response: Lessons from COVID-19 for Pandemic Health Equity. J Health Polit Policy Law. 2020 May; https://doi.org/10. 1215/03616878-8641469 PMID: 32464654
- 45. Gray DM, Anyane-Yeboa A, Balzora S, Issaka RB, May FP. COVID-19 and the other pandemic: populations made vulnerable by systemic inequity. Nature Reviews Gastroenterology and Hepatology. 2020. p. 520–2. https://doi.org/10.1038/s41575-020-0330-8 PMID: 32541960